Reviewer #1:
Scientific Quality: Grade C (Good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: Comparision between healthy covid 19 patient and heart disease patients with covid 19 infection.

Reviewer #2:
Scientific Quality: Grade B (Very good)
Language Quality: Grade A (Priority publishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: TOPIC IS GOOD WHY THERE IS NO CASES OR RESULTS TO MATCH AND EMPHASIZE ON THE STATED COVID-19 COMPLICATIONS.?

Authors:
Thank you both for your insightful comments. All requested additions have been made according to your suggestions as you may appreciate in the revised text. Moreover, a final thorough review regarding language quality has been made by Dr. Sonu Bhaskar who is a native English speaker